MEG Management Committee meeting

09.04.2008, 13:00hrs, West Wing Lecture Theatre, CBU

Minutes
Present:  Friedemann Pulvermuller (Chair), Rik Henson, Yury Shtyrov, Oleg Korzyukov, Clare Dine, Danny Mitchell (replacing Rhodri Cusack), Georgie Morrill (minutes).
1. Apologies for absences
· Rhodri Cusack
· Olaf Hauk
· Gary Chandler
· Rob Anthony
2. Matters arising 
Dan Wakeman has been approved to have operator status.
3. Minutes from the last meeting
RH reported an update to item 9 (AOB) which is that Jason Taylor has submitted an abstract to Biomag reporting his word versus non-word group localisation results.  FP expressed his interest in this study and RH informed that Jason will be showing results at the Wednesday Lunch Time Seminar.  YS, Jason Taylor, Dan Wakeman, Danny Mitchell, OK (using data from Detroit), and FP will be representing the CBU which is good news.

    The minutes were approved.
4. Elekta meeting









FP

FP reported that Elekta-Neuromag, EN, came back to us to ask if we could pay the outstanding 10% of the money that we owe.  The CBU declined this request because the system isn’t yet fully functional and mentioned work that needs to be done before the contract will be fulfilled. This includes the improvement of software packages and system functionality. EN agreed that there is need for further action before claims can be made.   YS pointed out that Anthea informed him that we have received a credit notice clearing the invoice.
5. GSK update










FP
FP informed the committee that he and YS had had a meeting with Dr. Pradeep Nathan and Soren Christensen to discuss a possible future project with GSK on schizophrenia.  The project would include a range of experiments (Memory, acoustic and cognitive MMN and P50).  RH suggested that OK attend any further GSK meetings regarding this as he is very familiar with schizophrenia studies.  FP agreed.  There will be 54 experimental subjects and 108 recordings. According to current plans, this new GSK project will use 4 2hour MEG slots per month from August 2008, increasing in number into 2009. The recordings for the project should be finished by the end of 2009, if possible earlier.
6. Computing
This is postponed till the next meet.
Action: 2nd workstation still needs to be moved to the lab – YS to coordinate with RA.

7. Lab Business and Progress

7.1  Sound insulation, progress – so progress can be easily made so we   should  try and get some reimbursement
7.2  Simultaneous MEG and EEG recordings – Discuss at next meeting.
7.3  MEG Operator / Researcher Qualification Procedure –Discuss at next meeting.
7.4  Eye tracker – Elekta confirmed that they will supply us with an SMI eye tracker.
7.5 Visual supine stimulation – Discuss at next meeting.
7.6  Foot pedal – Discuss at next meeting  
7.7  Other issues – The EEG caps are failing and the new cap seems to be     working okay but it was agreed to test it further before ordering a whole new set.  FP suggested ordering now but the operators would like to wait a bit because only 1 researcher has used the cap.  RH said that no one has checked the quality of the MEG data with the cap on.  CD and OK discussed the cleaning of the EEG caps and decided it is better to soak them for a while before cleaning as saves time in the long run. FP suggested ordering 3 of the most frequently used size.  YS suggested that we order 3 x sizes 58, 2 x 56 and 60 and 1 x 62 and 54.

Error reports were also discussed and OK/CD reported that the error messages only tend to happen twice a week now which is an improvement.  OK said that there were 2 bad channels which Elekta need to look at when they next visit.  FP suggested both early morning tuning and at 13:00 when there is the break in the scheduler.

FP reported that the HPI had caused problems with Max’s study and suggested all meeting to go over the method.  All members of the committee will meet at 2pm on Wednesday 23rd in the prep room to discuss and possibly agree on HPI methods.

RH asked whether the broken key on the keypad had been fixed.  CD and OK will look into this.
Action: GM to add technical failure data to stats spreadsheet, CD/OK/MvC to check broken key and label keys.  All to meet on Wednesday 23rd April at 2pm to go over HPI methods etc.
8. Lab Stats for March







GM showed the group the statistics for March and FP enquired about the decrease in usage of the lab.  GM explained that this was partly technical failure and partly due to maintenance afternoon that wasn’t needed and couldn’t be filled.  It was decided that GM should speak to GC and see if he is happy with just 1 afternoon every fortnight kept free for maintenance which can then be offered out to researchers a week or 2 before so it can still be used.  

Also the new scan booker system for the fMRI was mentioned and apparently it is easy to use and creates its own reports which would free up some of GM’s time.  GM will speak to Lucille about the system and see if she is happy to work with this.

Action:  GM to speak to Lucille and GC.
9. Lab cancellation policy

GM reported that there were a couple of incidents of very last minute cancellations meaning that the lab would not be used because it was too short notice to get a participant in.  The committee discussed this and decided that 10 calendar days would be appropriate for cancellations. Any slot cancelled after that period would count as a used slot.
10. AOB
Danny enquired about whether the de-magnetiser was ready.  YS said it was unlikely to achieve any progress on this but he would talk to GC.
11. Date of the next meeting
Wednesday 23rd April at 2pm in the MEG prep room (for HPI discussion). Wednesday 7th May at 2pm in the Lecture Theatre for the normal MEG MC meeting.
